Author's response to reviews

Title: Smoking, environmental tobacco smoke and risk of renal cell cancer: a population-based case-control study

Authors:

Ryan P. Theis (rtheis@ichp.ufl.edu)
Suzanne M. Dolwick Grieb (sdolwick@ufl.edu)
Deborah Burr (burr@stat.ufl.edu)
Tariq Siddiqui (siddit@medicine.ufl.edu)
Nabih R. Asal (asal@phhp.ufl.edu)

Version: 4 Date: 12 December 2008

Author's response to reviews:

Dear Editors,

We are pleased to hear that our manuscript has been accepted in principle for publication in BMC Cancer. Attached please find a revised version of our manuscript with the following minor changes:

- Page 16, 3rd sentence. The in-text citation originally included author's last name (Reynolds), which was not consistent with the manuscript's referencing format. "Reynolds" has been deleted from the in-text citation.
- Tables 2, 3, 4, 5, and 6. These have been converted to portrait layout since our original submission. Minor typographical corrections have been made to improve the layout of these tables and their overall readability.

No other changes have been made to the manuscript.

Sincerely,

Ryan Theis